

# Variations in blood pressure after a 75 g oral glucose load and their implications for detecting hypertension and postprandial hypotension in Chinese adults: a cross-sectional study

Xiaoying Zhou<sup>1†</sup>, Tongzhi Wu<sup>2,3†</sup>, Miaomiao Sang<sup>1,4</sup>, Shanhu Qiu<sup>5</sup>, Bei Wang <sup>6</sup>, Haijian Guo<sup>7</sup>, Kaili Li<sup>8</sup>, Qing Wang<sup>9</sup>, Xinling Wang<sup>10</sup>, Qingyun Chen<sup>11</sup>, Hong Li<sup>12</sup>, Sunjie Yan<sup>13</sup>, Michael Horowitz <sup>12</sup>, Christopher K. Rayner <sup>12</sup>, <sup>2,14</sup>, Duolao Wang<sup>15</sup>, Danny Liew<sup>16</sup>, Karen L. Jones <sup>12</sup>, <sup>2,3</sup>\*<sup>‡</sup>, and Zilin Sun<sup>1</sup>\*<sup>‡</sup>

<sup>1</sup>Department of Endocrinology, Zhongda Hospital, Institute of Diabetes, School of Medicine, Southeast University, 87 Ding Jia Qiao, Nanjing 210009, Jiangsu, China; <sup>2</sup>Centre of Research Excellence in Translating Nutritional Science to Good Health, Adelaide Medical School, University of Adelaide, AHMS Building, Cnr George St and North Tce, Adelaide, South Australia 5000, Australia; <sup>3</sup>Endocrine and Metabolic Unit, Royal Adelaide Hospital, Port Rd, Adelaide, South Australia 5000, Australia; <sup>4</sup>Department of Endocrinology and Genetic Metabolism, The First Affiliated Hospital of Wannan Medical College, Wuhu, Anhui, China; <sup>5</sup>Department of General Practice, Zhongda Hospital, Institute of Diabetes, School of Medicine, Southeast University, Nanjing, Jiangsu, China; <sup>6</sup>School of Public Health, Southeast University, Nanjing, Jiangsu, China; <sup>7</sup>Department of Integrated Services, Jiangsu Provincial Center for Disease Control and Prevention, Nanjing, Jiangsu, China; <sup>8</sup>Department of Endocrinology, Xinjiang Uygur Autonomous Region Hospital of traditional Chinese Medicine, Urumqi, Xinjiang Uyghur Autonomous Region, China; <sup>10</sup>Department of Endocrinology, First Affiliated Hospital of Jilin University, Rangenu, China; <sup>13</sup>Department of Guangxi Medical University, Nanjing, Jiangsu, China; <sup>14</sup>Department of Endocrinology, First Affiliated Hospital of Fujian Medical University, Usuangxi, China; <sup>15</sup>Department of Endocrinology, First Affiliated Hospital of Fujian Province, Fuzhou, Fujian, China; <sup>14</sup>Department of Gastroenterology and Hepatology, Agal Adelaide Hospital, Adelaide, South Australia, <sup>15</sup>Department of Clinical Sciences, Liverpool School of Tropical Medicine, Liverpool, United Kingdom; and <sup>16</sup>Adelaide Medical School, University of Adelaide, South Australia, Australia, Australia, Australia, Australia, Australia, Australia

Received 5 October 2024; revised 24 February 2025; accepted 13 March 2025; online publish-ahead-of-print 8 April 2025

Aims	Current hypertension guidelines fail to discriminate between fasting and postprandial blood pressure (BP) measurements. Meal ingestion often triggers a marked increase in splanchnic blood flow, potentially inducing a sustained fall in systolic BP of ≥20 mmHg, termed postprandial hypotension (PPH). This study aimed to evaluate BP responses to a 75 g glucose drink and its implications for detecting hypertension and PPH in community-dwelling adults.
Methods and results	A stratified multi-stage random sampling method was used to obtain a nationally representative sample of $n = 4429$ adult residents between April 2020 and January 2021 in China. BP and heart rate (HR) were measured before, and 1 and 2 h after, a 75 g glucose drink. When fasting, 38.4% of the study population had high BP (BP $\geq$ 140/90 mmHg). Following the glucose drink, SBP and DBP decreased (SBP by 6.2 [95% CI: 5.8, 6.6] mmHg and 8.1 [7.7, 8.5] mmHg, DBP by 4.7 [4.4, 4.9] mmHg and 6.1 [5.8, 6.4] mmHg), and HR increased (by 4.3 [4.0, 4.5] bpm and 2.6 [2.4, 2.9] bpm) at 1 and 2 h ( $P < 0.001$ for all), with only 30.9% and 27.0% of the study population having high BP at 1 and 2 h, respectively. After adjustment for age and sex distribution, 19.9% of the general population was estimated to have PPH. Postprandial hypotension was associated with an increased risk of combined cardiovascular disease and stroke.

\* Corresponding authors. Tel: +86 25 83262818, Email: sunzilin1963@126.com (Z.S.); Tel: +61 8 8313 7821, Email: karen.jones@adelaide.edu.au (K.L.J.)

 $^{\ddagger}$  K.L.J. and Z.S. are joint senior authors.

 $<sup>^{\</sup>dagger}$  X.Z. and T.W. are joint first authors.

<sup>©</sup> The Author(s) 2025. Published by Oxford University Press on behalf of the European Society of Cardiology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial License (https://creativecommons.org/licenses/by-nc/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited. For commercial re-use, please contact reprints@oup.com for reprints and translation rights for reprints. All other permissions can be obtained through our RightsLink service via the Permissions link on the article page on our site—for further information please contact journals.permissions@oup.com.

Conclusion	Ingestion of a 75 g glucose drink often lowers BP, frequently leading to PPH and influencing the detection of hypertension. Accordingly, guidelines for measurements of BP and interpretation of outcomes should consider the potential impact of meal ingestion on BP.
Lay summary	<ul> <li>Meal ingestion increases blood pooling into the gut, which may result in a sustained fall in systolic blood pressure (BP) and, in some cases, with a magnitude ≥20 mmHg, known as postprandial hypotension (PPH).</li> <li>This cross-sectional study showed in community-dwelling Chinese adults that, after a 75 g glucose drink, both systolic BP and diastolic BP decreased, and heart rate increased, with a reduced detection rate of hypertension and an estimated rate of PPH of 19.9%.</li> <li>Guidelines for measurements of BP and interpretation of outcomes should consider the potential impact of meal ingestion on BP.</li> </ul>

#### **Graphical Abstract**



## Introduction

Normalization of elevated blood pressure (BP) is pivotal in minimizing the risk of cardiovascular events, including stroke, and reducing mortality. For example, death from stroke, heart disease, and other vascular disorders approximately doubles with every 20 mmHg increase above 115 mmHg in systolic blood pressure (SBP) and every 10 mmHg increase above 85 mmHg in diastolic blood pressure (DBP),<sup>1</sup> while stringent control of BP, including the attainment of SBP within the range of 110–130 mmHg<sup>2</sup> or the maintenance of SBP within the target range of 120–140 mmHg over long term,<sup>3</sup> has been reported to substantially reduce cardiovascular events. Precise measurement of BP represents the initial step in the management of hypertension. Current guidelines for the diagnosis and management of hypertension provide recommendations relating to measurement procedures, and the potential impact of different settings on BP measurements (e.g. office vs. home), and emphasize the need to evaluate BP at specific times throughout the day and night, particularly with the use of ambulatory equipment.<sup>4–10</sup> However, while some guidelines suggest that home BP monitoring should be undertaken following an overnight fast, none provide specific advice about measurement of BP in relation to meals,<sup>4–10</sup> which potentially represents a major omission.

Meal ingestion triggers a series of haemodynamic responses, characterized by a marked increase in splanchnic blood flow, consequent reduction in peripheral blood volume, and the necessity for compensatory increases in cardiac output (reflected by increases in heart rate [HR]) and systemic vascular resistance to prevent a fall in BP.<sup>11</sup> If these compensatory mechanisms are impaired, postprandial hypotension (PPH), defined as a fall in SBP of  $\geq$ 20 mmHg within 2 h after the start of a meal or a glucose drink, or to <90 mmHg when baseline SBP is over 100 mmHg, may occur.<sup>12</sup> Hitherto, there is no consensus on the choice of a 'test meal' for the detection of PPH, but in many cases, the diagnostic 'meal' for PPH has been a glucose drink, as used in the oral glucose tolerance test for the diagnosis of type 2 diabetes and impaired glucose tolerance.<sup>11</sup> PPH has been shown to occur much more frequently than previously appreciated, with a high prevalence being reported in studies involving 'healthy' older people (~20%), hospitalized older patients (20-91%), nursing home residents (24-38%), individuals with hypertension (27.4-72.8%), and in long-standing type 1 and type 2 diabetes (~40%).<sup>11</sup> PPH also appears to occur more frequently than orthostatic hypotension, which is more widely recognized.<sup>13</sup> Importantly, PPH strongly predisposes to syncope, falls, coronary events, stroke, and mortality.<sup>14,15</sup> Evidence derived from cohort studies indicates that PPH is associated with an increased risk of cardiovascular disease (CVD) events as well as mortality in the elderly.<sup>16,17</sup> Moreover, a greater reduction in postprandial SBP is associated with a higher rate of both incident coronary events and total mortality.<sup>15</sup> A recent meta-analysis indicates that PPH is associated with a 52% increase in the risk of combined CVD outcomes and all-cause mortality.<sup>18</sup> These observations attest to the need to evaluate both fasting and postprandial BP in the diagnosis and management of hypertension.

To our knowledge, the BP response to a nutrient load and its determinants have not been evaluated in the general population. Moreover, the prevalence of PPH, its risk factors, and association with CVD have only been evaluated in specific populations with relatively small sample sizes.<sup>11,18</sup> The aim of this cross-sectional study was, accordingly, to determine the effects of a standardized nutrient load (75 g oral glucose) on BP and HR, as well as the prevalence and determinants of PPH, in a community-dwelling adult population.

## **Methods**

#### Study population

A cross-sectional study was conducted between April 2020 and January 2021 in China, which was embedded in the second follow-up of the SENSIBLE-cohort study (Study on Evaluation of iNnovated Screening tools and determination of optimal diagnostic cut-off points for type 2 diaBetes in Chinese muLti-Ethnic), designed to determine the optimal cut-off values for advanced glycation end-products and glycated haemoglobin (HbA1c) for diagnosing type 2 diabetes in China.<sup>19</sup> To select a nationally representative sample of adults in the general population, a multi-stage cluster and simple randomization method was used. In Stage 1, the sampling process was stratified according to geographic region (North, South, East, and West China), and six provinces, including Jilin (North), Yunnan and Guangxi (South), Fujian and Jiangsu (East), and Xinjiang Uyghur autonomous region (West), were selected. In Stage 2, cities in each province were numbered and selected (1-2 cities in each province) using a simple random sampling method, yielding a total of eight cities. In each city, 1 neighbourhood community and 1 administrative village were selected by a simple random sampling method. Accordingly, the study involved participants of six different ethnicities, including Han, Zhuang, Uygur, Kazak, Dai, and Korean. Finally, individuals who were  $\geq$ 18 years old and had lived in their current residence for  $\geq$ 5 years were eligible to participate and were stratified according to sex and age. Individuals who were pregnant, or had a history of diabetes, significant mental illness or a condition which precluded them from completing the study procedures, were excluded.

A total of 5000 subjects were invited, of whom 4511 participated, with an overall response rate of 90.2% (81.8% [1607/1964] for males and 95.7% [2904/3036] for females). After excluding 40 participants with missing data on age or sex and 42 participants with incomplete BP measurements due to technical problems, a total of 4429 participants were included in the final analysis (see Supplementary material online, *Figure S1*). The research protocol was approved by the Human Research Ethics Committee of Zhongda Hospital, Southeast University, Nanjing, China. All participants provided written informed consent.

#### Data collection

Participants were instructed to maintain their usual lifestyle for at least 3 days, and to fast overnight for a minimum of 10 h prior to being studied. Smoking was prohibited from the night before the study. Anti-hypertensive medications scheduled for the night before or morning of the study were withheld. On the study day, baseline BP and HR were measured on the non-dominant arm using a validated automated sphygmomanometer (YE900, Yuwell, China)<sup>20</sup> after participants had been seated in a quiet environment for at least 10 min (feet on floor, back supported). Vigorous exercise was avoided for at least 24 h before the study visit. The cuff size was chosen according to the arm circumference of each participant and was positioned on the upper nondominant arm at the level of the heart. A fasting venous blood sample was then obtained for measurements of HbA1c and fasting plasma glucose (FPG), triglycerides (TG), total cholesterol (TC), high-density lipoprotein (HDL) cholesterol, low-density lipoprotein (LDL) cholesterol, alanine aminotransferase (ALT), aspartate aminotransferase (AST), γ-glutamyl transpeptidase (GGT), total bilirubin (Tbil), uric acid (UA), creatinine (Cr), and blood urea nitrogen (BUN). A fasting spot urine sample was collected for measurements of albumin and creatinine. A standardized questionnaire was used to obtain information relating to sociodemographic characteristics and medical history. Body weight, height, waist circumference, and hip circumference were also measured. Body mass index (BMI) was calculated as body weight (kg) divided by height squared (m), and waist-hip ratio (WHR) was calculated as waist circumference (cm) divided by hip circumference (cm). Participants then ingested a 200 mL drink containing 75 g glucose. Measurements of BP and HR were repeated at 1 and 2 h after the drink. Due to logistical constraints, a single measurement was conducted in a consistent manner at each timepoint. A second venous blood sample was obtained for measurement of the 2h plasma glucose concentration (2h-PG). All the data were collected via standardized procedures by well-trained investigators.

All blood samples were processed on site within 30 min of collection, and conveyed by air in iceboxes to a central laboratory (Nanjing Adicon Clinical Laboratories) and analysed immediately upon arrival using high-performance liquid chromatography (D-10<sup>TM</sup>Hemoglobin Analyzer, Bio-Rad Inc., CA, USA) for HbA1c, and an automated chemistry analyzer (SynchronLX-20, Beckman Coulter Inc., CA, USA) for all other measurements.

#### Definitions

As previously, PPH was defined as a fall in SBP  $\geq$ 20 mmHg within 2 h after the glucose drink.<sup>12</sup> Both males and females were classified by BMI as lean (<24 kg/m<sup>2</sup>), overweight 24–27.9 kg/m<sup>2</sup>, or obese  $\geq$ 28 kg/m<sup>2,21</sup> Hypertension was defined as a baseline SBP ≥140 mmHg, diastolic DBP ≥90 mmHg, current use of anti-hypertensive medication, or previously diagnosed hypertension.<sup>8</sup> Glycaemic status was classified according to the 2020 Chinese Diabetes Society criteria:<sup>22</sup> diabetes was defined as a FPG  $\geq$ 7.0 mmol/L, 2h-PG  $\geq$ 11.1 mmol/L, HbA1c  $\geq$  6.5%, or a self-reported history of diabetes; prediabetes as impaired fasting glucose (FPG between 6.1 and 6.9 mmol/L) or impaired glucose tolerance (2h-PG between 7.8 and 11.0 mmol/L); and normoglycaemia as FPG <6.1 mmol/L and 2h-PG <7.8 mmol/L. The presence of macrovascular disease was determined by a self-reported history of CVD or stroke,<sup>23</sup> and micro-vascular disease by the presence of retinopathy (diagnosed according to the Early Treatment Diabetic Retinopathy Study criteria)<sup>24</sup> or microalbuminuria (diagnosed by a urine albumin: creatinine ratio > 30 mg/g).<sup>22</sup> Dyslipidaemia was defined as

Table T Demographic and biochemical parameters in the study population					
Characteristic	Overall	Male	Female		
	(n = 4429)	( <i>n</i> = 1547)	( <i>n</i> = 2882)		
Age (year)	56.8 ± 9.8	57.8 ± 10.0	56.3 ± 9.6		
BMI (kg/m <sup>2</sup> )	25.25 ± 3.83	25.26 ± 3.77	25.24 ± 3.86		
Fasting SBP (mm Hg)	135 ± 20	135 ± 20	135 ± 20		
Fasting DBP (mm Hg)	84 <u>±</u> 12	85 ± 12	83 <u>±</u> 12		
Fasting HR (bpm)	73 <u>+</u> 12	72 ± 12	73 <u>+</u> 12		
FPG (mmol/L)	5.35 <u>±</u> 0.78	5.40 ± 0.78	5.32 ± 0.79		
2h-PG (mmol/L)	7.35 ± 2.36	7.34 ± 2.40	7.35 ± 2.34		
HbA1c (%)	5.63 ± 0.67	5.64 ± 0.76	5.62 ± 0.61		
TC (mmol/L)	4.69 ± 0.95	4.61 ± 0.92	4.74 <u>+</u> 0.96		
TG (mmol/L)	1.26 (0.92, 1.84)	1.24 (0.90, 1.81)	1.23 (0.87, 1.74)		
HDL-c (mmol/L)	$1.42 \pm 0.28$	1.39 ± 0.29	1.44 <u>+</u> 0.28		
LDL-c (mmol/L)	$2.62 \pm 0.70$	$2.58 \pm 0.68$	2.65 ± 0.70		
UA (mmol/L)	297 <u>±</u> 84	313 ± 84	288 <u>+</u> 82		
Cr (umol/L)	57 <u>±</u> 14	$60 \pm 14$	56 <u>+</u> 13		
eGFR (mL/min/1.73 m <sup>2</sup> )	103.6 ± 12.6	103.5 ± 12.3	103.6 ± 12.8		
BUN (umol/L)	$5.1 \pm 1.4$	$5.2 \pm 1.4$	5.0 ± 1.4		
ALT (U/L)	15 (10, 21)	15 (11, 23)	14 (10, 21)		
AST (U/L)	21 (18, 26)	22 (18, 26)	21 (18, 26)		
GGT (U/L)	19 (13, 29)	20 (14, 34)	18 (13, 27)		
Tbil (umol/L)	10.0 (7.7, 13.2)	10.6 (8.1, 14.4)	9.7 (7.5, 12.7)		
Hypertension (n [%])	1982 (44.8)	713 (46.1)	1269 (44.0)		
Diabetes (n [%])	545 (12.3)	211 (13.6)	334 (11.6)		
Dyslipidaemia (n [%])	950 (21.4)	355 (22.9)	595 (20.6)		
CKD (n [%])	26 (0.6)	8 (0.5)	18 (0.6)		
Hepatic dysfunction ( <i>n</i> [%])	109 (2.5)	51 (3.3)	58 (2.0)		

 Table 1
 Demographic and biochemical parameters in the study population<sup>a</sup>

<sup>a</sup>Some values were missing, including n = 18 for body mass index (BMI), n = 17 for fasting heart rate (HR), n = 64 for 2 h plasma glucose (2h-PG), n = 9 for glycated haemoglobin (HbA1c), n = 2 for fasting plasma glucose (FPG), total cholesterol (TC), triglycerides (TG), high-density lipoprotein cholesterol (HDL-c), low-density lipoprotein cholesterol (LDL-c), uric acid (UA), estimated glomerular filtration rate (eGFR) and blood urea nitrogen (BUN), n = 4 for alanine aminotransferase (ALT), aspartate aminotransferase (AST),  $\gamma$ -glutamyl transpeptidase (GGT), and total bilirubin (Tbil).

either high TC (≥6.20 mmol/L), high TG (≥2.30 mmol/L), low HDL cholesterol (<1.00 mmol/L), or high LDL cholesterol (≥4.10 mmol/L); chronic kidney disease was defined as a reduced eGFR (<60 mL/min/1.73m<sup>2</sup>), calculated by the CKD-EPI formula;<sup>25</sup> hepatic dysfunction was defined as an elevation of liver enzymes at least twice the upper limit of the normal reference range (ALT ≥80 U/L, AST ≥100 U/L, GGT ≥120 U/L, Tbil ≥40 µmol/L).<sup>26</sup>

#### Statistical analysis

The sample size was estimated on the basis of the cross-sectional design of the study. If the estimated prevalence of PPH in the general population is approximately 20%, with the maximum allowable error at 2% and the design effect at 2, a minimum of n = 3073 participants would be required. To allow for a response rate of 90%, the minimum sample size would be 3414. Categorical data are presented as percentages, and continuous data as mean ± standard deviation or median (25th-75th percentile). Repeatedmeasures ANOVA was used to compare BP and HR responses to oral glucose. Factors associated with changes in BP and HR following the glucose load were analyzed using mixed linear models, with time as the fixed effect, subject as random effect, and age, sex, and ethnicity as co-variates. Subjects were further stratified according to age (below and above the median age of 58 years), sex, and microalbuminuria status to evaluate factors associated with changes in SBP, DBP, and HR. The overall prevalence of PPH was adjusted for age and sex according to the 2020 Chinese population census.<sup>27</sup> Estimates of the prevalence of PPH in age groups <40 years, 40-49 years, 50-59 years, 60–69 years, and ≥70 years were weighted according to sex distribution, and estimates stratified by sex were adjusted according to the population proportions in age groups <40 years, 40-49 years, 50-59 years, 60-69 years, and ≥70 years. Crude rates of PPH were first stratified by BMI (lean, overweight, or obese), hypertension (present or not), and glycaemic status (normoglycaemia, prediabetes, or diabetes). Because PPH is frequently a reflection of an inadequate compensatory increase in HR, we also examined the associations between the changes in SBP and HR at both 1 and 2 h after oral glucose in participants with and without PPH using Pearson correlation analysis. Logistic regression analyses were then used to assess the association of PPH with macrovascular and micro-vascular disease, with adjustment for age, sex, ethnicity, BMI, hypertension, HbA1c, TC, TG, and eGFR. Because the number of missing values for co-variates considered in the regression analysis was modest (<10%), substitution of missing values was not conducted in the analyses. A two-sided P-value of <0.05 was considered significant. All statistical analyses were conducted using SPSS 25.0 (Chicago, IL, USA).

# Results

#### Characteristics of study population

Table 1 shows the characteristics of the study population which comprised n = 4429 participants (34.9% male, median age 56.8 ± 9.8 years, range 19–87 years) of 6 ethnicities (n = 3336 Han, n = 196 Zhuang, n =225 Uygur, n = 285 Kazak, n = 193 Dai, and n = 194 Korean) sampled

Vital signs	Statistics	Fasting	1 h	2 h	Р
SBP (mmHg)	n	4429	4429	4312	
	Mean $\pm$ SD	135 ± 20	129 <u>+</u> 19	127 <u>+</u> 18	<0.001
	Difference between 1 h	and fasting (95% CI) <sup>a</sup>		-6.2 (-6	.6, –5.8)
	Difference between 2 h	and fasting (95% CI) <sup>b</sup>		-8.1 (-8	.5, –7.7)
DBP (mmHg)	n	4429	4429	4312	
	Mean $\pm$ SD	84 ± 12	79 <u>+</u> 11	78 <u>+</u> 11	<0.001
	Difference between 1 h	and fasting (95% CI) <sup>a</sup>		-4.7 (-4	.9, -4.4)
	Difference between 2 h	and fasting (95% CI) <sup>b</sup>		-6.1 (-6	.4, -5.8)
HR (bpm)	n	4412	4419	4305	
	Mean $\pm$ SD	73 ± 12	77 ± 12	76 <u>+</u> 11	<0.001
	Difference between 1 h	and fasting (95% CI) <sup>a</sup>		4.3 (4.0	0, 4.5)
	Difference between 2 h	and fasting (95% CI) <sup>b</sup>		2.6 (2.4	4, 2.9)
SBP/DBP	≥140/90	38.4%	30.7%	26.9%	< 0.001
	120-140/80-90	41.8%	38.7%	39.4%	<0.001
	<120/80	19.8%	30.4%	33.7%	<0.001

Table 2	Blood pressure and heart rate measurements a	it baseline and 60 and	120 min following o	oral glucose (75 g	)
---------	--	------------------------	---------------------	--------------------	---

SBP, systolic blood pressure; DBP, diastolic blood pressure; HR, heart rate.

<sup>a</sup>Calculated by 1 h minus fasting levels.

<sup>b</sup>Calculated by 2 h minus fasting levels.

from 6 provinces (n = 2867 Jiangsu, n = 87 Fujian, n = 196 Guangxi, n = 809 Xinjiang Uyghur autonomous region, (n = 195 Yunan, and n = 275 Jilin). Overall, 44.8% (n = 1982) participants had a previously diagnosed hypertension or high fasting BP recorded on the study day, 12.3% newly diagnosed diabetes, 21.4% dyslipidaemia, 0.6% chronic kidney disease, and 2.5% hepatic dysfunction. Compared with males, female participants were slightly younger, had lower UA levels, but comparable BMI, BP, and glycaemic and lipid profiles.

#### Systolic blood pressure, diastolic blood pressure, and heart rate before and after a 75 g oral glucose load

The 75 g oral glucose load was associated with reductions in both SBP (by 6.2 [95% CI: 5.8, 6.6] mmHg and 8.1 [95% CI: 7.7, 8.5] mmHg at 1 and 2 h, respectively) and DBP (by 4.7 [95% CI: 4.4, 4.9] mmHg and 6.1 [95% CI: 5.8, 6.4] mmHg at 1 and 2 h, respectively), and an increase in HR (by 4.3 [95% CI 4.0, 4.5] bpm and 2.6 [95% CI: 2.4, 2.9] bpm at 1 and 2 h, respectively) (P < 0.001 for all) (*Table 2*, Supplementary material online, *Figure S2*).

The proportions of participants with high BP (i.e.  $\geq$  140/90 mmHg) were 38.4% when fasting (n = 1700/4429), 30.7% (n = 1360/4429) at 1 h, and 26.9% (n = 1161/4312) at 2 h after oral glucose. In 28.6% (n = 1266/4429) of individuals, BP was either high while fasting and normal at 2 h after the oral glucose load (n = 872), or was normal before and high after glucose (n = 394).

In males, high BP occurred in 38.6% when fasting, 31.7% at 1 h, and 28.0% at 2 h. In females, high BP occurred in 38.3% when fasting, 30.2% at 1 h, and 26.4% at 2 h (see Supplementary material online, *Table S1*).

In the subgroup with pre-existing hypertension, high BP occurred in 85.8% when fasting, 59.3% at 1 h, and 53.0% at 2 h (see Supplementary material online, *Table S1*).

In the subgroup with microalbuminuria, high BP occurred in 60.1% when fasting, 50.6% at 1 h, and 45.1% at 2 h. In the subgroup without

microalbuminuria, high BP occurred in 33.3% when fasting, 26.0% at 1 h, and 22.6% at 2 h (see Supplementary material online, *Table S1*).

# Factors associated with changes in blood pressure and heart rate after oral glucose

After adjustment for potential confounders, including age, sex and ethnicity, changes from fasting SBP and DBP at 1 or 2 h were inversely associated with age, fasting SBP and DBP, hypertension, dyslipidaemia, and microalbuminuria, while changes in HR were greater in males than females, and associated directly with BMI and dyslipidaemia, but inversely with age, fasting BP, fasting HR, and hypertension (Table 3). Stratification of subjects according to age (below and above the median 58 years) yielded similar outcomes to those overall, with the exception of significant interactions between age, fasting SBP (P = 0.038), and hypertension (P = 0.002) in relation to the changes in DBP (see Supplementary material online, Table S2). However, the magnitude of the difference between subjects aged <58 years and  $\geq 58$  years was marginal. Stratification of subjects according to sex showed no interaction between sex and the aforementioned factors to influence the BP and HR responses to oral glucose (see Supplementary material online, Table S3). Stratification of subjects according to microalbuminuria status revealed a significant interaction between the presence of microalbuminuria and diabetes in relation to changes in SBP (coefficient -2.77,95% CI: [-5.41, -0.12], P = 0.040), such that the reduction in SBP after oral glucose was greater among individuals with microalbuminuria and diabetes (see Supplementary material online, Table S4). Stratification of subjects according to a prior history of hypertension revealed significant interactions between hypertension and age (coefficient -0.06, 95% CI: [-0.10, -0.01], P = 0.010) and dyslipidaemia (coefficient -1.10, 95% CI: [-2.15, -0.06], P = 0.039) in relation to changes in DBP. However, the magnitude of differences between subgroups with and without a prior history of hypertension were marginal (see Supplementary material online, Table S5). Association analyses of the changes in SBP and HR at 1 and 2 h after oral glucose also revealed

		Coefficient (95% CI), P-value	
Characteristics	∆SBP	ΔDBP	ΔHR
Sex (male)	0.66 (-0.08, 1.40), 0.081	-0.27 (-0.74, 0.20), 0.261	0.77 (0.25, 1.30), 0.004
Age (year)	-0.09 (-0.13, -0.06), < 0.001	-0.04 (-0.07, -0.02), < 0.001	-0.04 (-0.07, -0.01), 0.003
BMI (kg/m <sup>2</sup> )	-0.02 (-0.12, 0.07), 0.631	-0.03 (-0.09, 0.03), 0.376	0.11 (0.04, 0.17), 0.002
Fasting SBP	-0.30 (-0.31, -0.28), < 0.001	-0.07 (-0.08, -0.06), < 0.001	-0.03 (-0.04, -0.01), < 0.001
Fasting DBP	-0.20 (-0.23, -0.18), < 0.001	-0.30 (-0.32, -0.29), < 0.001	-0.04 (-0.06, -0.02), < 0.001
Fasting HR	-0.02 (-0.05, 0.01), 0.146	-0.04 (-0.06, -0.02), < 0.001	-0.33 (-0.35, -0.31), < 0.001
Hypertension	-8.51 (-9.18, -7.85), < 0.001	-4.23 (-4.66, -3.80), < 0.001	-0.87 (-1.38, -0.36), 0.001
Diabetes	1.09 (-0.07, 2.26), 0.065	0.35 (-0.39, 1.09), 0.357	-0.55 (-1.38, 0.27), 0.190
Dyslipidaemia	-0.97 (-1.83, -0.11), 0.027	-0.63 (-1.18, -0.09), 0.022	0.50 (-0.11, 1.11), 0.107
Chronic kidney disease	-2.89 (-7.53, 1.74), 0.222	0.98 (-1.96, 3.91), 0.513	-2.14 (-5.44, 1.15), 0.202
Microalbuminuria	-2.08 (-3.00, -1.16), < 0.001	-0.95 (-1.53, -0.37), 0.001	-0.51 (-1.16, 0.15), 0.130
Hepatic dysfunction	1.72 (-0.54, 3.99), 0.136	1.42 (-0.02, 2.85), 0.052	0.51 (-1.10, 2.13), 0.533

 Table 3
 Factors associated with changes in blood pressure and heart rate following oral glucose (75 g)<sup>a</sup>

SBP, systolic blood pressure; DBP, diastolic blood pressure; HR, heart rate; BMI, body mass index.

Dyslipidaemia: Elevated TC, TG, LDL-c, or decreased HDL-c.

Chronic kidney disease: Decreased eGFR (lower than 60).

Hepatic dysfunction: Elevated ALT, AST, GGT, or Tbil (2 times higher than upper limit).

Microalbuminuria: Urine albumin to creatinine ratio over 30 mg/g.

<sup>a</sup>Adjusted for age, sex, and ethnicity.

significant correlations between the two parameters in participants without PPH (1 h: r = 0.097, P < 0.001; 2 h: r = 0.090, P < 0.001), but not in those with PPH (1 h: r = 0.042, P = 0.165; 2 h: r = 0.059, P = 0.053).

#### Prevalence of postprandial hypotension and its association with macro- and micro-vascular disease

The overall prevalence of PPH in the was 24.9% (n = 1101/4429). After adjustment for age and sex distribution according to the 2020 Chinese population census data, the estimated prevalence of PPH in the general Chinese adult population was 19.9% (95% CI: 18.7, 21.1%), with a higher prevalence in females than males (21.8% [95% CI: 20.3, 23.3%] vs. 18.0% [95% CI: 16.1, 20.0%], P = 0.003) and an increasing prevalence with ageing (from 12.5% [95% CI: 8.6, 16.3%] in participants <40 years to 27.6% [95% CI: 22.3, 32.8%] in those  $\geq$  70 years, P < 0.001) (*Table 4*). The prevalence of PPH was higher in overweight and obese than lean participants (26.2% [95% Cl: 23.4, 29.1%] in obese, 26.2% [95% Cl: 24.2, 28.3%] in overweight, and 22.9% [95% CI: 20.9, 24.8%] in lean participants, P = 0.04), and in those with hypertension (39.8% [95% Cl: 37.6, 31.9%] vs. 12.8% [95% CI: 11.5, 14.1%], P < 0.001) and prediabetes (normal glucose tolerance 23.3% [95% Cl: 21.8, 24.9%], prediabetes 29.2% [95% Cl: 26.7, 32.0%], diabetes 23.7% [95% Cl: 19.8, 27.6%], P < 0.001) than those without (Table 4). Stratification of participants according to fasting SBP, cigarette smoking, alcohol intake, and the use of anti-hypertensive medications further showed that the prevalence of PPH was higher among individuals with higher fasting SBP, cigarette smoking, and alcohol intake, but was not affected by the use of antihypertensive medications (see Supplementary material online, Table S6).

After adjustment for age, sex, ethnicity, hypertension, glycaemic state, CKD, and microalbuminuria, female sex (OR 1.26, 95% Cl 1.08, 1.47), age (per 5-year increment, OR 1.05, 95% Cl 1.01, 1.10),

hypertension (OR 4.55, 95% CI 3.89, 5.33), and CKD (OR 2.88, 95% CI 1.19, 6.97) were identified as independent risk factors for PPH. Surprisingly, diabetes appeared to be associated with a reduced risk of PPH (OR 0.72, 95% CI 0.56, 0.93) (*Table 5*).

The prevalence of CVD, stroke, and combined macrovascular disease were substantially higher in those with PPH than those without (CVD 6.3% vs. 3.9%, stroke 3.1% vs. 1.7%, combined 9.2% vs. 5.1%, all P < 0.05); after adjustment for age, sex, ethnicity, BMI, hypertension, HbA1c, TC, TG, and eGFR, PPH was associated with a higher risk of combined CVD and stroke (OR 1.49, 95% CI 1.13–1.97) (*Table 6*). While the prevalence of retinopathy and microalbuminuria were also higher in participants with PPH, after adjustment for sociodemographic and metabolic confounders, PPH was no longer associated with microvascular disease.

## Discussion

This large, nationwide study, involving community-dwelling adult participants from different geographic regions and ethnic backgrounds in China, has demonstrated a high prevalence of a marked fall in BP following a 75 g oral glucose load. This phenomenon would have a significant impact on the diagnosis of hypertension, given that about a third of individuals with an abnormally high BP level at baseline were normotensive during the 2 h after the glucose drink. Moreover, about 25% of the study population (equating to ~20% of the standardized Chinese adult population) exhibited a fall in SBP of greater than 20 mmHg from baseline after oral glucose, meeting the widely used diagnostic criterion for PPH. The latter was associated with both increasing age and hypertension, and also predictive of a marked increase in macrovascular disease. These observations are indicative of major limitations in current practice relating to BP measurement and interpretation of BP outcomes.

Current guidelines for BP measurement generally lack information about the impact of meal ingestion, although some recommend that

Table 4

Characteristic	No of participants <sup>a</sup>	No of patients with PPH <sup>a</sup>	Prevalence (%)	P value <sup>b</sup>
Overall	4429	1101	19.9 (18.7, 21.1) <sup>c</sup>	
Sex				
Male	1547	362	18.0 (16.1, 20.0) <sup>c</sup>	0.003
Female	2882	739	21.8 (20.3, 23.3) <sup>c</sup>	
Age (years)				
19–39	289	39	12.5 (8.6, 16.3) <sup>c</sup>	<0.001
40–49	602	130	21.4 (18.1, 24.7) <sup>c</sup>	
50–59	1710	443	25.3 (23.2, 27.3) <sup>c</sup>	
60–69	1545	411	26.5 (24.3, 28.7) <sup>c</sup>	
≥70	283	78	27.6 (22.3, 32.8) <sup>c</sup>	
BMI (kg/m <sup>2</sup> )				
<24.0	1750	400	22.9 (20.9, 24.8)	0.040
24.0–27.9	1758	461	26.2 (24.2, 28.3)	
≥28.0	903	237	26.2 (23.4, 29.1)	
Hypertension				
No	2447	313	12.8 (11.5, 14.1)	<0.001
Yes	1982	788	39.8 (37.6, 41.9)	
Glycaemic status				
Normal glucose tolerance	2848	664	23.3 (21.8, 24.9)	<0.001
Prediabetes	1125	329	29.2 (26.6, 31.9)	

108

Prevalence of postprandial hypotension stratified by demographic characteristics and health status

<sup>a</sup>Unweighted number of participants and PPH cases.

Diabetes

Table 5

<sup>b</sup>Prevalence between groups were compared using  $\chi^2$  test. Data are presented as percentage and 95% CI.

<sup>c</sup>Estimated prevalence of PPH in the general population, adjusted for age and/or sex of the latest Chinese population census data.

456

nypotension	
Characteristics	Odds ratio (95% Cl), P-value
••••••	
Female	1.26 (1.08, 1.47), 0.004
Age (per 5 year increment)	1.05 (1.01, 1.10), 0.011
Hypertension	4.55 (3.89, 5.33), <0.001
Glycaemic state	
Normal glucose tolerance	Reference
Prediabetes	1.06 (0.90, 1.26), 0.478
Diabetes	0.72 (0.56, 0.93), <0.001
Chronic kidney disease	2.88 (1.19, 6.97), 0.019
Microalbuminuria	0.97 (0.81, 1.17), 0.974

Factors associated with postprandial

<sup>a</sup>Logistic regression model adjusted for age, sex, ethnicity, hypertension, glycemic state, CKD, and microalbuminuria

BP is measured before breakfast and ingestion of medications for home BP monitoring.<sup>4,6,8,9</sup> Our study establishes that there is a major discrepancy in the detection of an abnormally high BP among males and females, individuals with a prior history of hypertension, and those with and without microalbuminuria, depending on whether BP is measured before or after oral glucose, reflecting the hypotensive response to the latter.

We have shown previously that the rate of gastric emptying, which varies substantially even between healthy individuals in the range of 1-4 kcal/min,<sup>28</sup> is a major determinant of the hypotensive response

# Table 6Association of postprandial hypotension withcomorbidities

23.7 (19.8, 27.6)

Vascular disease	Prevalence in non-PPH (%)	Prevalence in PPH (%)	Adjusted odds ratio (95% CI) of PPH <sup>a</sup>
Macrovascular disease	171/3325 (5.1)	99/1076 (9.2) <sup>b</sup>	1.49 (1.13, 1.97)
CVD	129/3325 (3.9)	68/1072 (6.3) <sup>b</sup>	1.31 (0.95, 1.82)
Stroke	57/3325 (1.7)	33/1074 (3.1) <sup>b</sup>	1.51 (0.95, 2.39)
Micro-vascular disease	693/3321 (20.9)	293/1099 (26.7) <sup>b</sup>	0.96 (0.81, 1.14)
Retinopathy	192/3211 (6.0)	65/1064 (6.1)	0.87 (0.64, 1.19)
Microalbuminuria	550/3169 (17.4)	253/1070 (23.6) <sup>b</sup>	0.98 (0.81, 1.17)

<sup>a</sup>Multivariable logistic regression adjusted for age, sex, ethnicity, body mass index (BMI), hypertension, glycated haemoglobin (HbA1c), total cholesterol (TC), triglycerides (TG), and estimated globular filtration rate (eGFR).

<sup>b</sup>P < 0.05 compared with the prevalence in non-PPH using  $\chi^2$  test.

to oral glucose.<sup>29</sup> Although a glucose drink differs from a physiological meal, previous work has shown that carbohydrate, fat, and proteins have comparable effects to reduce BP.<sup>30,31</sup> Moreover, the rate of gastric emptying, which is a dominant determinant of the BP response to a meal<sup>32</sup> is dependent on caloric content, rather than macronutrient composition.<sup>30</sup> Accordingly, we believe that our findings are of relevance to the ingestion of physiological meals so that measurement of BP should incorporate standardization of meal ingestion, involving both fasting and postprandial measurements—otherwise there is a high risk of misinterpreting BP results measured at convenience. For example, individuals who had abnormally high BP during fasting may be normotensive or even hypotensive, after a meal. Furthermore, the fasting and postprandial phases have not been discriminated in relation to the effect of antihypertensive drugs. We acknowledge that the diagnostic cut-off for PPH is somewhat arbitrary, attesting to the relative lack of clinical attention to this condition. While the current study was not designed to address this issue, we found that PPH diagnosed by a fall in SBP of  $\geq$ 20 mmHg after a 75 g oral glucose load was associated with increased vascular comorbidities. Moreover, that the use of a glucose drink for the detection of PPH can be readily combined with measurement of plasma glucose in screening for diabetes, adds to the diagnostic value.

After nutrient intake, blood is redistributed to the gut leading to compensatory increases in HR and systemic resistance, which are governed by autonomic function. When cardiovascular compensation is inadequate, BP falls. Indeed, in participants without PPH, the changes in SBP at 1 and 2 h after oral glucose correlated with the corresponding changes in HR, but these correlations were not evident in participants with PPH. As in previous studies,<sup>14,15,33</sup> we found that age and fasting BP were positively associated with the decline in postprandial BP, and negatively associated with the increase in HR, implying that in older individuals and those with hypertension the rise in HR was insufficient to maintain BP at fasting levels. The mechanisms underlying these associations remain to be defined, but may relate, at least in part, to autonomic dysfunction, which we did not assess, and the attenuated gastrovascular reflex seen in the elderly and in those with hypertension.<sup>34</sup> We recognize that vagal stimulation can result from glucose ingestion and may potentially drive a fall in BP, but this is typically accompanied by a concurrent fall in HR, rather than the observed increase.<sup>35</sup> Accordingly, vagal stimulation is most unlikely to account for the fall in BP evident in our study.

Based on the outcomes of our study, it can be estimated that about 20% of the total Chinese adult population would have PPH after a 75 g oral glucose load. Previous studies in relatively small samples indicated that PPH occurs commonly in the elderly, those with hypertension, and in individuals with long-standing type 1 or type 2 diabetes, <sup>14,17,32,36</sup> but the current study is the first to show that PPH occurs frequently in the general adult population. We also found that PPH was associated with a 51% increase in the combined risk of CVD and stroke, even after adjusting for age, hypertension, HbA1c and lipid levels. Our observations extend insights derived from prospective cohort studies in low-level care residents and older nursing home residents, which indicate that PPH represents an independent risk factor for the onset of coronary events, stroke, and overall mortality,<sup>15,33</sup> and a recently published meta-analysis which showed that PPH was associated with increased total CVD and stroke, as well as all-cause mortality and CVD mortality.<sup>18</sup> Accordingly, PPH is likely to represent an important target for preventing and managing macrovascular and micro-vascular disease and warrants evaluation in prospective studies. The fact that management options for PPH are currently limited, and suboptimal, must also be addressed.

In interpreting our findings, several limitations should be noted. First, females and older individuals were oversampled, as males and younger residents were often working outside their permanent residential regions. To account for this, we adjusted the estimated prevalence of PPH in the general community for the distribution of sex and age according to the 2020 national population census. Moreover, the

observed prevalence of diabetes and hypertension were comparable to those reported in other large-scale population-based studies (diabetes 12.8%, hypertension 44.7%),<sup>37,38</sup> supporting the representative nature of our study sample. It should be noted that people with known diabetes were excluded from the study and diabetes represented an incidental, rather than established, diagnosis. It was, accordingly, to be expected that the number of participants with diabetes was more modest when compared to those with normoglycaemia or prediabetes. The apparently reduced risk of PPH in the context of diabetes was surprising and likely to represent the outcome of these confounders. Second, individuals with known diabetes, mental illness, and other disorders that would prevent them from completing the study procedures, as well as those who were pregnant, were excluded, with the inherent potential for selection bias. Third, BP was not measured continuously for logistical reasons, potentially leading to an under-reporting of the prevalence of PPH. While it is unclear whether the falls in BP would be more marked after lunch and evening meals, there is evidence that the fall is usually greatest after breakfast,<sup>39</sup> and variations in BP after breakfast are more closely associated with cardiovascular mortality.<sup>17</sup> Twenty-four-hour ambulatory BP monitoring would be ideal for evaluating overall postprandial BP changes and for the detection of PPH but was not feasible in a study of this scale. Similarly, for logistical reasons, BP measurements were only performed on the study day, rather than on three different days, as suggested by current guidelines. This would affect the accuracy of the diagnosis of hypertension. Furthermore, for logistical reasons, BP was only measured once at each time point, which would inevitably increase data variability. Participants were instructed to withhold their antihypertensive medications only the night before or on the morning of the study, as we considered that more prolonged withholding of these medications would be unethical. This approach may, however, have accounted for the rise in BP in a small subset of participants. Fourth, our study lacked a control group who drank only water, so a gradual reduction in anxiety, rather than glucose intake, may potentially have affected BP levels. However, in previous studies which used water as a control, BP levels were still lower after oral glucose.<sup>40</sup> Importantly, the reduction in BP after oral glucose was accompanied by persistent increases in HR. Accordingly, it is highly unlikely that the fall in BP was driven by 'resting' or ease of anxiety. We recognise that the resting phase before each BP measurement, albeit complying with many guidelines,<sup>4,8–10</sup> was relatively brief, but prolongation of the resting phase would have compromised the feasibility of this study. While there may have been inaccuracy in BP measurements generally, this would have led to random data misclassification—that is, not biased towards any subgroup of interest—and, hence, conservative estimates of differences among them. Future studies, involving a non-nutritive drink as a control, would add support on the impact of a nutrient load on BP levels. Fifth, this was a cross-sectional study with selfreported macrovascular diseases, so that reporting bias cannot be excluded. Since evaluation of BP and HR responses was conducted along with screening of diabetes, diabetic micro-vascular complications, including diabetic retinopathy and diabetic kidney disease, were used as a surrogate of micro-vascular events. It should also be noted that in our study, the potential determinants of changes in BP and HR were exploratory, and could be compromised by inadequate adjustment for other unknown confounders. Longitudinal studies are needed to confirm the consequences of PPH in the general adult population. Finally, the current study was conducted only in a Chinese adult population. In view of substantial ethnic disparities in the regulation of metabolic homeostasis, further studies to evaluate the prevalence and consequences of PPH in other populations are warranted.

# Conclusion

In conclusion, our study showed that there is a substantial decrease in BP after oral glucose in a Chinese community-dwelling adult population, with 20% meeting the criteria for PPH. Our observations highlight the need for standardization of BP measurement in relation to meals, and indicate that PPH, in addition to hypertension, represents a major target for clinical intervention to prevent CVD and stroke.

# Supplementary material

Supplementary material is available at European Journal of Preventive Cardiology.

## **Author contributions**

Z.S., K.L.J., S.Q., B.W., M.H., C.K.R., T.W., D.L., and D.W. conceived and designed the study. B.W., H.G., K.L., Q.W., X.W., Q.C., H.L., S.Y., X.Z., and M.S. conducted the epidemiological survey. X.Z., M.S., S.Q., and DW performed the statistical analysis. All authors contributed to acquisition, analysis, or interpretation of data. X.Z. and T.W. drafted the manuscript. All authors revised the report and approved the final version before submission.

#### Funding

We acknowledge grant funding from the Key Research and Development Program in Jiangsu Province (BE2022828), National Key Research and Development Program of China (2016YFC1305700), and The Hospital Research Foundation.

# Ethical approval and consent to participate

The study was approved by the Human Research Ethics Committee of Zhongda Hospital, Southeast University, Nanjing, China. All participants provided written informed consent.

**Conflict of interest:** M.H. has participated in the advisory boards and/or symposia for Novo Nordisk, Sanofi, Novartis, Eli Lilly, Merck Sharp & Dohme, Boehringer Ingelheim, and AstraZeneca and has received honoraria for this activity. C.K.R. has received research funding from AstraZeneca, Merck Sharp & Dohme, Eli Lilly, Novartis, and Sanofi. T.W. has received travel support from Novartis and Sanofi and research funding from Novartis and AstraZeneca. None of the other authors has any personal or financial conflict of interest to declare.

#### Data availability

Relevant individual de-identified participant data will be made available on reasonable request via e-mail to Z.S.

#### References

- Lewington S, Clarke R, Qizilbash N, Peto R, Collins R. Age-specific relevance of usual blood pressure to vascular mortality: a meta-analysis of individual data for one million adults in 61 prospective studies. *Lancet* 2002;**360**:1903–1913.
- Deng Y, Bai J, Yang X, Liu W, Guo Z, Zhang J, et al. Achieved systolic blood pressure and cardiovascular outcomes in 60–80-year-old patients: the Strategy of Blood Pressure Intervention in the Elderly Hypertensive Patients (STEP) trial. *Eur J Prev Cardiol* 2023; 30:1017–1027.
- Lin Z, Xiao Z, Chen W, Xu W, Huang C, Xie J, et al. Association of long-term time in target range for systolic blood pressure with cardiovascular risk in the elderly: a Chinese veteran cohort study. Eur J Prev Cardiol 2023;30:969–977.

a

- 4. Whelton PK, Carey RM, Aronow WS, Casey DE Jr, Collins KJ, Dennison Himmelfarb C, et al. 2017 ACC/AHA/AAPA/ABC/ACPM/AGS/APhA/ASH/ASPC/NMA/PCNA Guideline for the Prevention, Detection, Evaluation, and Management of High Blood Pressure in Adults: A Report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines. *Hypertension* 2018;**71**:e13–e115.
- Rabi DM, McBrien KA, Sapir-Pichhadze R, Nakhla M, Ahmed SB, Dumanski SM, et al. Hypertension Canada's 2020 comprehensive guidelines for the prevention, diagnosis, risk assessment, and treatment of hypertension in adults and children. Can J Cardiol 2020;36:596–624.
- Umemura S, Arima H, Arima S, Asayama K, Dohi Y, Hirooka Y, et al. The Japanese Society of Hypertension Guidelines for the Management of Hypertension (JSH 2019). *Hypertens Res* 2019;42:1235–1481.
- Hypertension in Adults: Diagnosis and Management. London: National Institute for Health and Care Excellence (NICE); 2023.
- 8. Writing Group of 2018 Chinese Guidelines for the Management of Hypertension, Chinese Hypertension League, Hypertension Branch of China International Exchange and Promotive Association for Medical and Health Care, Hypertension Branch of Chinese Geriatrics Society, Hypertension Branch of China Association of Geriatric Health and Care, Chinese Stroke Association, et al. 2024 Chinese guidelines for the management of hypertension. *Chin J Hypertens* 2024;**32**:603–700.
- 9. Kim HL, Lee EM, Ahn SY, Kim KI, Kim HC, Kim JH, et al. The 2022 focused update of the 2018 Korean Hypertension Society Guidelines for the management of hypertension. *Clin Hypertens* 2023;**29**:11.
- Mancia G, Kreutz R, Brunström M, Burnier M, Grassi G, Januszewicz A, et al. 2023 ESH Guidelines for the management of arterial hypertension The Task Force for the management of arterial hypertension of the European Society of Hypertension: Endorsed by the International Society of Hypertension (ISH) and the European Renal Association (ERA). J Hypertens 2023;41:1874–2071.
- Trahair LG, Horowitz M, Jones KL. Postprandial hypotension: a systematic review. J Am Med Dir Assoc 2014;15:394–409.
- Jansen RW, Lipsitz LA. Postprandial hypotension: epidemiology, pathophysiology, and clinical management. Ann Intern Med 1995;122:286–295.
- Vloet LC, Pel-Little RE, Jansen PA, Jansen RW. High prevalence of postprandial and orthostatic hypotension among geriatric patients admitted to Dutch hospitals. J Gerontol A Biol Sci Med Sci 2005;60:1271–1277.
- Aronow WS, Ahn C. Postprandial hypotension in 499 elderly persons in a long-term health care facility. J Am Geriatr Soc 1994;42:930–932.
- Aronow WS, Ahn C. Association of postprandial hypotension with incidence of falls, syncope, coronary events, stroke, and total mortality at 29-month follow-up in 499 older nursing home residents. J Am Geriatr Soc 1997;45:1051–1053.
- Jang A. Postprandial hypotension as a risk factor for the development of new cardiovascular disease: a prospective cohort study with 36 month follow-up in community-dwelling elderly people. J Clin Med 2020;9:345.
- Zanasi A, Tincani E, Evandri V, Giovanardi P, Bertolotti M, Rioli G. Meal-induced blood pressure variation and cardiovascular mortality in ambulatory hypertensive elderly patients: preliminary results. J Hypertens 2012;30:2125–2132.
- Jenkins DJA, Sahye-Pudaruth S, Khodabandehlou K, Liang F, Kasmani M, Wanyan J, et al. Systematic review and meta-analysis examining the relationship between postprandial hypotension, cardiovascular events, and all-cause mortality. Am J Clin Nutr 2022;**116**: 663–671.
- Liu Y, Sang M, Yuan Y, Du Z, Li W, Hu H, et al. Novel clusters of newly-diagnosed type 2 diabetes and their association with diabetic retinopathy: a 3-year follow-up study. Acta Diabetol 2022;59:827–835.
- Zhang HJ, Zhang J, Wang SL, Zhang J, Teng LN, Zhang SJ, et al. Validation of the YuWell YE900 oscillometric blood pressure monitor for professional office use in adults and children according to the AAMI/ESH/ISO Universal Standard (ISO 81060-2:2018). Blood Press Monit 2021;26:396–399.
- Chen C, Lu FC. The guidelines for prevention and control of overweight and obesity in Chinese adults. *Biomed Environ Sci* 2004;**17**:1–36.
- Chinese Diabetes Society. Guideline for the prevention and treatment of type 2 diabetes mellitus in China (2020 edition). *Chin J Diabetes Mellitus* 2021;13:315–409.
- World Health Organization (WHO). Cardiovascular Diseases (CVDs) https://www. who.int/news-room/fact-sheets/detail/cardiovascular-diseases-(cvds) (26 July 2024).
- Klein R, Knudtson MD, Lee KE, Gangnon R, Klein BE. The Wisconsin Epidemiologic Study of Diabetic Retinopathy XXIII: the twenty-five-year incidence of macular edema in persons with type 1 diabetes. *Ophthalmology* 2009;**116**:497–503.
- KDIGO. 2024 clinical practice guideline for the evaluation and management of chronic kidney disease. *Kidney Int* 2024;**105**:S117–S314.
- Kwo PY, Cohen SM, Lim JK. ACG clinical guideline: evaluation of abnormal liver chemistries. Am J Gastroenterol 2017;112:18–35.
- National Bureau of Statistics of China. 2021 China Statistical Yearbook. 2021. https:// www.stats.gov.cn/sj/pcsj/rkpc/7rp/indexce.htm (26 December 2021).

- Phillips LK, Deane AM, Jones KL, Rayner CK, Horowitz M. Gastric emptying and glycaemia in health and diabetes mellitus. *Nat Rev Endocrinol* 2015;**11**:112–128.
- Vanis L, Gentilcore D, Hausken T, Pilichiewicz AN, Lange K, Rayner CK, et al. Effects of gastric distension on blood pressure and superior mesenteric artery blood flow responses to intraduodenal glucose in healthy older subjects. Am J Physiol Regul Integr Comp Physiol 2010;299:R960–R967.
- Gentilcore D, Hausken T, Meyer JH, Chapman IM, Horowitz M, Jones KL. Effects of intraduodenal glucose, fat, and protein on blood pressure, heart rate, and splanchnic blood flow in healthy older subjects. Am J Clin Nutr 2008;87:156–161.
- Oberoi A, Giezenaar C, Rigda RS, Horowitz M, Jones KL, Chapman I, et al. Effects of co-ingesting glucose and whey protein on blood glucose, plasma insulin and glucagon concentrations, and gastric emptying, in older men with and without type 2 diabetes. *Diabetes Obes Metab* 2023;25:1321–1330.
- Jones KL, Tonkin A, Horowitz M, Wishart JM, Carney BI, Guha S, et al. Rate of gastric emptying is a determinant of postprandial hypotension in non-insulin-dependent diabetes mellitus. *Clin Sci (Lond)* 1998;**94**:65–70.
- Fisher AA, Davis MW, Srikusalanukul W, Budge MM. Postprandial hypotension predicts all-cause mortality in older, low-level care residents. J Am Geriatr Soc 2005;53: 1313–1320.

- van Orshoven NP, Oey PL, van Schelven LJ, Roelofs JM, Jansen PA, Akkermans LM. Effect of gastric distension on cardiovascular parameters: gastrovascular reflex is attenuated in the elderly. J Physiol 2004;555:573–583.
- Gierthmuehlen M, Plachta DT. Effect of selective vagal nerve stimulation on blood pressure, heart rate and respiratory rate in rats under metoprolol medication. *Hypertens Res* 2016;**39**:79–87.
- Jones KL, MacIntosh C, Su YC, Wells F, Chapman IM, Tonkin A, et al. Guar gum reduces postprandial hypotension in older people. J Am Geriatr Soc 2001;49:162–167.
- Lu J, Lu Y, Wang X, Li X, Linderman GC, Wu C, et al. Prevalence, awareness, treatment, and control of hypertension in China: data from 1.7 million adults in a population-based screening study (China PEACE Million Persons Project). *Lancet* 2017;**390**:2549–2558.
- Li Y, Teng D, Shi X, Qin G, Qin Y, Quan H, et al. Prevalence of diabetes recorded in mainland China using 2018 diagnostic criteria from the American Diabetes Association: national cross sectional study. BMJ 2020;369:m997.
- Vloet LC, Smits R, Jansen RW. The effect of meals at different mealtimes on blood pressure and symptoms in geriatric patients with postprandial hypotension. J Gerontol A Biol Sci Med Sci 2003;58:1031–1035.
- Trahair LG, Rajendran S, Visvanathan R, Chapman M, Stadler D, Horowitz M, et al. Comparative effects of glucose and water drinks on blood pressure and cardiac function in older subjects with and without postprandial hypotension. *Physiol Rep* 2017;5:e13341.